Author’s response to reviews

Title: Sex differences in the prevalence and adverse outcomes of sarcopenia and sarcopenic obesity in community dwelling elderly in east China using the AWGS criteria

Authors:

Yanping Du (yanpingdu@fudan.ceu.cn)
Xiaodong Wang (wangxiaodongting@163.com)
Hua Xie (xiehua_dr@163.com)
Songbai Zheng (zhengsongbai@126.com)
Xiaoqing Wu (13816845504@163.com)
Xiaoying Zhu (zxiaoying909@sina.com)
Xuemei Zhang (zhangxuemei@163.com)
Sihong Xue (xuesihong@126.com)
Huilin Li (sirius1121@hotmail.com)
Wei Hong (hongweibaby@126.com)
Wenjing Tang (twj1028@163.com)
Minmin Chen (cmm86916@163.com)
Qun Cheng (quncheng2016@163.com)
Jianqin Sun (jianqins@163.com)

Version: 2 Date: 22 Jul 2019

Author’s response to reviews:

Editor Comments:

Pay particular attention to the objections raised by reviewer 2

Answer: The article has been revised according to the objections raised by reviewer 2.
Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Answer: We use the reputable English language editing service from American Journal Experts (http://bit.ly/AJE_BS) to improve the language in this article. (Certificate Verification Key: E2AE-0B8F-171F-28CE-50AA)